Author’s response to reviews

Title: The information and support needs of patients discharged after a short hospital stay for treatment of low-risk Community Acquired Pneumonia: implications for treatment without admission.

Authors:

Deborah J Baldie (d.baldie@dundee.ac.uk)
Vikki A Entwistle (V.A.Entwistle@dundee.ac.uk)
Peter G Davey (p.g.davey@chs.dundee.ac.uk)

Version: 6 Date: 25 July 2008

Author’s response to reviews:

there were only a small number of changes that were typographical in nature e.e. incorrect use if italics , consistent use of abbreviations throughout document.